Author's response to reviews

Title: Clinical Features and Risk factors for Severe and Critical Pregnant Women with 2009 Pandemic H1N1 Influenza Infection in China

Authors:

Peng-jun Zhang (15801669400@163.com)
Xiao-li Li (5162202@163.com)
Bin Cao (caobin1999@gmail.com)
Shi-gui Yang (ysgcdc@163.com)
Li-rong Liang (llrcruie@yahoo.com.cn)
Li Gu (guliangel@yahoo.com.cn)
Zhen Xu (xuzhen@chinacdc.cn)
Ke Hu (huke-rmhospital@163.com)
Hong-yuan Zhang (zhly2008@yahoo.cn)
Xi-xin Yan (yanxixin@hotmail.com)
Wen-bao Huang (hub11281225@163.com)
Wei Chen (188006918@qq.com)
Jing-xiao Zhang (zjingxiao1101@163.com)
Lan-juan Li (lijii@zju.edu.cn)
Chen Wang (cyh-birm@263.net)

Version: 6 Date: 29 December 2011

Author's response to reviews: see over
Dear Jonathan A McCullers and Tonilynn Manibo,

Thank you for your comments to our manuscript. We are very grateful to have a chance to revise it. As requested we have dealt with the good comments of editor and present these below:

Point-by-point responses to the editorial comments:

Response to Editor:

1. Abstract, Results, line 7, change "and 38 were succeeded." to "and 38 were successful with NIV."
   Response: Thanks for your advice. We have changed "and 38 were succeeded." to "and 38 were successful with NIV."

2. p. 5, Introduction, line 13, delete the comma in the sentence "information is, limited..."
   Response: Thanks for your suggestion. We have deleted the comma.

3. p. 6, Methods, line 18, change "was" to "were" in the sentence "dysfunction or unconsciousness was contraindications..."
   Response: Thanks for your advice. We have changed "was" to "were".

4. p. 6, Methods, line 22, change "down ring" to "during"
   Response: Thanks for your reminder. We have changed "down ring" to "during".

5. p. 7, Methods, line 1, change "was" to "were" in the sentence "The data was entered..."
   Response: Thanks for your advice. We have changed "was" to "were".

6. p. 7, Methods, line 13-15, change sentence "And then the factors with statistical significance (p < 0.05) in the univariate analyses, then these factors were included the multivariate logistic regression analysis." to "Factors with statistical significance (p < 0.05) in the univariate analyses were included in the multivariate logistic regression analysis."
   Response: Thanks for your advice. We have changed it as you suggested.

7. p. 12, Conclusions, line 7, change "3) Premature delivery was independent risk factors for neonatal death." to "3) Premature delivery was an independent risk factor for neonatal death."
   Response: Thanks for your reminder. We have added the "an".

8. Delete the "Key Messages" Section as it is redundant with the "Conclusions" paragraph immediately preceding it.
   Response: Thanks for your suggestions. We have deleted the "Key Messages" Section.

9. Figure 1, lower right box, change "Neonatal" (both occurrences) to "Neonates"
   Response: Thanks for your suggestions. We have changed "Neonatal" to "Neonates".